Author's response to reviews

Title: A Randomised Controlled Trial to Explore Attitudes to Routine Scale and Polish and Compare Manual Versus Ultrasonic Scaling

Authors:

Brian C Bonner (b.c.bonner@dundee.ac.uk)
Linda Young (l.v.young@dundee.ac.uk)
Patricia A Smith (p.a.smith@dundee.ac.uk)
Wendy McCombes (w.mccombes@btopenworld.com)
Janet E Clarkson (j.e.clarkson@dundee.ac.uk)

Version: 2 Date: 20 May 2005

Author's response to reviews: see over
In the UK, dental care is provide either privately or within the National Health Service at a fixed cost that includes a patient contribution of 80% (up to a maximum charge for a course of treatment). Around 49% of the population are registered with an NHS dentist at any one time [4] but, as registration lapses if a visit is not made within 15 months, a greater proportion of the population is believed to receive care within the NHS dental service. Although the number of dental treatments provided outside the NHS scheme is increasing in the UK, an estimated 81% of treatments are currently provided under the NHS system [5] and it is likely that the majority of the population will continue to seek NHS dental care. All the patient participants in this study were receiving a scale and polish under the NHS system, for which a charge of £8.08 was in place at the time of the study.